Author's response to reviews

Title: A cluster-randomized trial to determine the impact of hotspot-targeted interventions on malaria transmission

Authors:

Teun Bousema (teun.bousema@lshtm.ac.uk)
Jennifer Stevenson (jennifer.stevenson@lshtm.ac.uk)
Amrish Baidjoe (amrish.baidjoe@gmail.com)
Gillian Stresman (gillian.stresman@lshtm.ac.uk)
Jamie T Griffin (immo.kleinschmidt@lshtm.ac.uk)
Immo Kleinschmidt (jamie.griffin@imperial.ac.uk)
Edmond J Remarque (remarque@bprc.nl)
John Vulule (jvulule@kemricdc.org)
Nabie Bayoh (nbayoh@kemricdc.org)
Kayla Laserson (klaserson@ke.cdc.gov)
Meghna Desai (mdesai@ke.cdc.gov)
Robert Sauerwein (r.sauerwein@mmb.umcn.nl)
Chris Drakeley (chris.drakeley@lshtm.ac.uk)
Jonathan Cox (jonathan.cox@lshtm.ac.uk)

Version: 2 Date: 27 August 2012

Author's response to reviews:

I have uploaded the three ethics approvals, a sponsor letter and the front page of a grant agreement with the Bill & Melinda Gates Foundation (999,997 USD).

I can confirm that we are currently recruiting participants and have a design heading in the main text.

Kindly let me know if there are additional questions.

yours sincerely,
Teun Bousema